Author's response to reviews

Title: Impact of changes in specificity of data recording on cause-specific injury mortality in the United States, 1999-2010

Authors:

Guoqing Hu (hguoqing009@gmail.com)
Keita Mamady (kethmohawassdeh@gmail.com)

Version: 2
Date: 21 June 2014

Author's response to reviews:

Dear Editor,

Many thanks for your email to request us to resubmit the manuscript!

As you required, we add the statement of competing interest and authors' contribution after the references.

As for the error of number of authors in the cover letter, we are sorry for it. It is a mistake, the number of authors is two for this paper. I don't know how to correct the error in the submitted version on June 18, 2014 online.

Best wishes,

Guoqing Hu

June 21, 2014